To
Editor-in-Chief
World Journal of Clinical Oncology
Sub: Resubmission of an article

We would like to submit the revised article entitled “The Bethesda system for reporting thyroid cytopathology: A three-year study at a tertiary care referral center in Saudi Arabia” (Manuscript NO: 31202). We express our gratitude for the reviewers’ valuable positive comments. Since, the reviewer’s not requested to do any modification in the article we are submitting the same manuscript for your kind perusal. We are looking forward for your consideration to the above-mentioned article for publication at the World Journal of Clinical Oncology.

Kind regards

Mohamed Abdulaziz Al Dawish
Department of Endocrinology and Diabetes,
Diabetes Treatment Center,
Prince Sultan Military Medical City,
Alrabwa. P.O Box 261370
Riyadh, 11342,
Kingdom of Saudi Arabia
Phone 00966-1- 477-7714 ext 40504
Fax 00966-12912224

Email: maldawish@psmmc.med.sa

COMMENTS TO AUTHORS
Question: Dear Authors, I have read with interest your manuscript entitled “The Bethesda system for reporting thyroid cytopathology: a three-year study at a tertiary care referral center in Saudi Arabia” submitted to the World Journal of Clinical Oncology. The study shows a very exhaustive analysis of the throughput of thyroid cytopathology over a three-year period. The manuscript contains a detailed exposition of the results, including comprehensive tables and a comparison to other recent studies. In my opinion, this manuscript fulfills all the requirements to be published.

Answer: Many thanks for your appreciative comments.

COMMENTS TO AUTHORS
This paper introduced the condition on diagnosing thyroid nodules using FNB in Saudi. The material and result were good, and the discussion is apt to the title.

Answer: Many thanks for your appreciative comments.